Author’s response to reviews

Title: Sex knowledge, attitudes, and high-risk sexual behaviors among unmarried youth in Hong Kong

Authors:

Paul S.F. Yip (sfpyip@hku.hk)  
Huiping Zhang (zhp0205@gmail.com)  
T.H. Lam (hrmrth@hku.hk)  
Kwok Fai Lam (hrntlkf@hku.hk)  
A M Lee (amlee@hkucc.hku.hk)  
Susan Fan (sfan@famplan.org.hk)

Version: 5 Date: 19 June 2013

Author’s response to reviews: see over
Author’s response to reviews

Title: Sex knowledge, attitudes, and high-risk sexual behaviors among unmarried youth in Hong Kong

Authors: Paul SF Yip (sfpyip@hku.hk)
Huiping Zhang (zhp0205@gmail.com)
T.H Lam (hrmrlth@hku.hk)
Kwok Fai Lam (hrntlkf@hku.hk)
A. Lee (amlee@hkucc.hku.hk)
S. Fan (sfan@famplan.org.hk)

Version: 4 Date: 19 June 2013

Author’s response to reviews: see over
Reviewer's report

Title: Sex knowledge, attitudes, and high-risk sexual behaviors among unmarried youth in Hong Kong

Version: 4 Date: 29 May 2013
Reviewer: Ronald Thomas

Reviewer's report:
After extensive review of the authors revisions, I firmly believe that they have more than adequately addressed all previous concerns in an exemplary fashion specific to the areas of major, minor, and discretionary issues previously brought forth by all reviewers. It is my opinion that this article is well written, the revisions have strengthened what was an already strong manuscript, and one which has major importance to the readership.

Level of interest: An article of outstanding merit and interest in its field

Quality of written English: Acceptable

Statistical review: Yes, and I have assessed the statistics in my report.

Declaration of competing interests:
'I declare that I have no competing interests'.

Reviewer's report

Title: Sex knowledge, attitudes, and high-risk sexual behaviors among unmarried youth in Hong Kong

Version: 4 Date: 4 May 2013
Reviewer: qiaoqin ma

Reviewer's report:
Minor revision:
1. What does “change of pregnancy” mean in “measure” section?

To address your concern, we need to clarify that it is “chances of pregnancy” in our measure section instead of “change of pregnancy”.

2. In discussion section, "Half of the sexually active youth who did not use contraceptive
methods indicated that sexual intercourse happened unexpectedly, so they had no time to plan the contraception issues." I think that the author is better not to arbitrarily make such conclusion, is it possible that some men many not care about their female partner's pregnancy?

As suggested, we have included your explanation into our argument on page 13 as follow:

“It is also possible that some men may not care about their female partner’s pregnancy”.

**Level of interest:** An article whose findings are important to those with closely related research interests

**Quality of written English:** Acceptable

**Statistical review:** No, the manuscript does not need to be seen by a statistician.

**Declaration of competing interests:**
I have no competing interest in relation to this paper.